Author’s response to reviews

Title: What is the Difference in Morphologic Features of the Lumbar Vertebrae Between Caucasian and Taiwanese Subjects? A CT-based Study: Implication of Pedicle Screws Placement via Roy-Camille or Weinstein Methods

Authors:

Hsi-Hsien Lin (hsihsienlin@gmail.com)

Jung-Pan Wang (jpwang0801@gmail.com)

Cheng-Li Lin (jengli94@gmail.com)

Yu Cheng Yao (orthycyao@gmail.com)

Shih-Tien Wang (stwang@vghtpe.gov.tw)

Ming-Chau Chang (mcchang@vghtpe.gov.tw)

Po-Hsin Chou (choupohsin@gmail.com)

Version: 5 Date: 09 Nov 2018

Author’s response to reviews:

Dear Editor-In-Chief:

Some tables were not elucidated in the word text. Accordingly, please refer to supplementary material for the letter of Response to the Reviewers. Thanks a lot. I am really sorry for the inconvenience.

Best regards,

Po-Hsin Chou